\begin{table}%t1 
  \caption{\label{datos_radio}Radio data compiled from the literature for our WR~galaxy sample. } 
  \small%\centering
\par
  \begin{tabular}{l ccc cc}
  \hline\hline\noalign{\smallskip}
 Galaxy  &  $F_{\rm H~ I}$  & $W_{\rm H~ I}$ & Ref.  &  $S_{\rm 1.4~ GHz}$  &  Ref.     \\
         & [Jy km s$^{-1}$]  & [km s$^{-1}$]   &      &     [mJy]         &          \\ 
    \hline
HCG 31 AC      &   5.15        &  169.2A+190.6C    & VM05    &   22~$\pm$~3       & VM05 \\ 
HCG 31 B       &   2.74        &    85.8           & VM05    &   2.1~$\pm$~0.3    & VM05 \\
HCG 31 F       &   0.866       &    74.6           & VM05    &   \nodata        & \nodata \\ 
HCG 31 G       &   2.74        &    84.9           & VM05    &   3.3~$\pm$~0.5    & VM05 \\ 
Mkn 1087       &   5.38        &    270     & GG81 &   12.1~$\pm$~0.6     & Co98 \\
Haro 15        &  3.11~$\pm$~1.01&    220     & GG81 &  17.8~$\pm$~1.0  & Co98\\ 
Mkn 1199       &   1.78~$\pm$~0.67 &    170     & DC04 &   36.2~$\pm$~1.2     & Co98\\
Mkn 5          &  2.12~$\pm$~0.27&22.4~$\pm$~4.9& Pa03 &  $<$2.8     & HSLD02 \\
IRAS 08208+2816& \nodata       & \nodata & \nodata &   15.2~$\pm$~0.6  & Co98 \\
IRAS 08339+6517& 3.68~$\pm$~0.46 &  $\sim$300 & Ca04 &  33.56 & Co90  \\   
POX 4       &  4.31         &  130    & LS10b &   4.2~$\pm$~0.5  & Co98\\  
UM 420     &    \nodata  & \nodata & \nodata     &   1.1~$\pm$~0.3  &  HSLD02 \\ 
SBS 0926+606A  & 1.30~$\pm$~0.49 & 120~$\pm$~37 & P02  & 2.7~$\pm$~0.6   & HSLD02  \\   
SBS 0926+606B  & 1.10~$\pm$~0.49 & 120~$\pm$~37 & P02  & \nodata     & \nodata  \\   
SBS 0948+532   & \nodata  & \nodata & \nodata      & $<$0.9  &  HSLD02  \\
SBS 1054+365   & 4.03~$\pm$~0.39 & 117~$\pm$~11 & Z00   &   1.28~$\pm$~0.14 & BWH95 \\ 
SBS 1211+540   & 0.71~$\pm$~0.12  & 47 & H05      & $<$0.9  &  HSLD02   \\
SBS 1319+579   & 8.4  & 134     & H07      & $<$2.9  &  HSLD02   \\
SBS 1415+437   & 4.73~$\pm$~0.32 &   66      &  H05  & $<$0.5  &  H05 \\ 
III Zw 107 &  4.48~$\pm$~0.79     &   200~$\pm$~25    & P03 & 8.0~$\pm$~0.5  & Co98+Y01 \\
Tol 9        &  5.02~$\pm$~0.40 &   185     &  LS10b & 19.2~$\pm$~0.7  & Co98  \\         
Tol 1457-262  &  4.3    &    176       &  Kor06 & 38.9~$\pm$~1.8 & Co98+Y01 \\   
Arp 252  &   \nodata  & \nodata & Kor06     & 97.6~$\pm$~3.0  & Co98+Y01\\  
NGC 5253   &  42.1~$\pm$~2.3  &  95~$\pm$~3  &  LS10a & 78~$\pm$~3 & LS10a \\   
\hline
  \end{tabular}
  \tablebib{Ca04:  Cannon et~al. (\cite{Cannon2004}); 
  CBG04:  Casasola et~al. (\cite{Casasola2004}) 
  Co90:  Condon et~al. (\cite{Condon1990}); 
  Co98:  Condon et~al. (\cite{Condon1998}); 
  DC04:  Davoust \& Contini (\cite{Davoust2004}); 
  GG81:  Gordon \& Gottesman (\cite{Gordon1981});  
  HSLD02:  Hopkins et~al. (\cite{Hopkins2002}); 
  H05:  Huchtmeier et~al.  (\cite{Huchtmeier2005}); 
  H07:  Huchtmeier et~al. (\cite{Huchtmeier2007});
  Kor06: Koribalski (2006), priv. comm.; 
  LS10a:  \mbox{L\'opez-S\'anchez} et~al. (\cite{opez2010ae});
  LS10b:  \mbox{L\'opez-S\'anchez} et~al. (\cite{opez2010be});
  Pa03:  Paturel et~al. (\cite{Paturel2003}): HyperLEDA; 
  P02:  Pustilnik et~al. (\cite{Pustilnik2002});   
  VM05:  \mbox{Verdes-Montenegro} et~al. (\cite{Verdes2005}); 
  Y01:   Yun et~al. (\cite{Yun2001}); 
  Z00:  Zasov et~al. (\cite{Zasov2000}).}
  %\vspace*{6mm}
\end{table}